Fall 1994-1995 CS/ECE 752 Readings Part 1 Prof. Mark D. Hill

Regarding History, Performance, and Instruction Sets [BGv46] A. W. BURKS, H. H. GOLDSTINE and J. VON NEUMANN, “Preliminary discussion of the logical design of an electronic computing instrument”,Report to the U.S. Army Ordinance Department (1946). [Cas92] B. CASE, “Updated SPEC Benchmarks Released”, Microprocessor Report, 6, 12 (16 September 1992), 14-19. [Wul81] W. A. WULF, “Compilersand Computer Architecture”, Computer, 14, 7 (July 1981), 41-48. [BhC91] D. BHANDARKAR and D. W. CLARK, “Performance from Architecture: Comparing a RISC and CISC with Similar Hardware Organization”, Proc. Fourth International Conference on Architectural Support for Programming Languages and Operating Systems, Palo Alto (April 1991), 310-319. [SmW94] J. E. SMITH and S. WEISS, “PowerPC 601 and Alpha 21064: A Tale of Two RISCs”, IEEE Computer , 27 , 6 (June 1994), 12-24.